Dear Editor,

Thank you very much for giving us the opportunity to revise our manuscript. We appreciate the editors and reviewers for their valuable and constructive comments and suggestions on our manuscript entitled “Clustering cases of Chlamydia Psittaci Pneumonia Mimicking COVID-19 Pneumonia” (Manuscript NO.: 69959).

We have uploaded the revised files according to the requirement of the journal, which we hope will be met with approval.

Sincerely,

Jun Liu

Department of Radiology, The Second Xiangya Hospital, Central South University, Changsha, Hunan 410011, China

Department of Radiology Quality Control Center, Hunan Province, Changsha, Hunan 410011, China

Tel: +86-13787085002

E-mail: 19918944475@163.com, Luohong1003@163.com, junliu123@csu.edu.cn